MIT Course 18.06 Linear Algebra: Spring 2003

Lectures: Johan de Jong MWF at 11:00 room 54-100

Web page: http://web.mit.edu/18.06/www

Text: Introduction to Linear Algebra (Second Edition) by Gilbert Strang

Videotapes: On webpage and in Engineering Library

Recitations: You must be firmly enrolled in a specific section (instructor and time) because your homeworks and exams will go to that section. Changes of section are made in the Undergraduate Math Office (2-108). Your recitation instructor – not the lecturer – is the person to ask about homework problems and grades, either in recitation or office hours.

Homework: Assignments will be due on Wednesdays, BEFORE 4:00PM. Please put them in your section’s box in 2-106, next to the Undergraduate Mathematics Office (2-108). Please staple them (you may use the UMO stapler). They are due every week except exam weeks and returned in recitation. The homeworks are essential in learning linear algebra. You are encouraged to talk to other students about the problems. But you must write up your own solutions. No late homeworks accepted.

Exams: There will be two one hour in class exams. They will be on Monday March 3rd and Wednesday April 9th. MIT has a procedure to deal with no shows on these exams. Please contact your recitation instructor (or the UMO) by phone or email on the morning of the exam if you cannot make it to the exam. You will have to go the the office of the counseling dean 5-104 to be excused from the exam.

There is also a 3 hour final exam. Similar procedures for absenses hold here.

Grading: Problem sets 30%, two one-hour exams 30%, final exam 40%.

# Time Room Instructor Office Phone Email @math Lec MWF 11 54-100 A.J. de Jong 2-270 Recitations

1 M 2 2-131 O. Kovrijkine 2-273 3-4380 oleg 2 M 3 2-131 C. Boulet 2-333 3-7826 cilanne 3 M 3 2-132 W-L. Yee 2-251 3-7566 wlyee 4 T 11 2-132 I. Pak 2-390 3-4390 pak 5 T 12 2-132 I. Pak 2-390 3-4390 pak 6 T 1 2-131 A. Malkin 2-180 3-4350 malkin 7 T 1 2-132 J. Farley 2-384 3-1798 8 T 2 2-132 J. Farley 2-384 3-1798 9 T 2 2-131 A. Malkin 2-180 3-4350 malkin Course Administrator: C. Boulet